Author's response to reviews

Title: Recruitment of aged donor heart with pharmacological stress echo. A case report.

Authors:
- Giorgio Arpesella (arpesella@aosp.bo.it)
- Sonia Gherardi (soniagherardi@libero.it)
- Tonino Bombardini (tbombardini@yahoo.it)
- Eugenio Picano (picano@ifc.cnr.it)

Version: 4 Date: 19 January 2006

Author's response to reviews:

Dear Miss Gemma Burns,
I resubmitted my manuscript final version on January 12th 2006. The formatting changes I made are the following:
References - In the manuscript text, I placed the references before the full stops.
Figures - I uploaded the additional files changing their names
I changed some verbs and words and explained better some concepts according to reviewers' requests.
With all best wishes
Dr. Tonino Bombardini